Author's response to reviews

Title: Benzyl butyl phthalate induces migration, invasion, and angiogenesis of Huh7 hepatocellular carcinoma cells through nongenomic AhR/G-protein signaling

Authors:

Cheng-Fang Tsai (chenfung2001@yahoo.com.tw)
Tsung-Hua Hsieh (pelagice@yahoo.com.tw)
Jau-Nan Lee (jaunanlee@hotmail.com)
Chia-Yi Hsu (husonweihsu@hotmail.com)
Yu-Chih Wang (joelovefu@gmail.com)
Feng-Jie Lai (D950819@mail.chimei.org.tw)
Kung-Kai Kuo (kuoksfo@yahoo.com.tw)
Hua-Lin Wu (halnwu@mail.ncku.edu.tw)
Eing-Mei Tsai (tsaieing@yahoo.com)
Po-Lin Kuo (kuopolin@seed.net)

Version: 6
Date: 14 July 2014

Author's response to reviews: see over
Dear Editor:

Greetings of Peace!

I am pleased to return my revised manuscript, and I have carefully checked and retyped the paper according to the checklist you sent me previously. It would very much appreciate if the referees can read the article soon, and I hope to hear from you again in this regard. Thank you for your time and attention to this matter.

Sincerely yours,

Eing-Mei Tsai
Department of Obstetrics and Gynecology Kaohsiung Medical University Hospital, Kaohsiung Medical University Hospital, Kaohsiung Medical University, Kaohsiung, No. 100, Shih-Chuan 1st Road, Kaohsiung 807, Taiwan.
Tel: 886-7-312-1101 ext. 6424
Fax: 886-7-321-0701
Email: tsaieing@yahoo.com
1) Please clarify whether the experiments involving animals have been specifically approved by an Ethics committee, and include the full name of the committee and a reference number as appropriate

The experiments involving animals have been specifically approved by by Kaohsiung Medical University Hospital Institutional Animal Care and Use Committee (IACUC).

IACUC Approval No: 101060

2) Copy-edit

We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

We appreciate the reviewers’ suggestion and have used a professional language editing service recommended by BioMed Central to improve the style of written English. The parts that changed have been marked in red in the manuscript.